Author's response to reviews

Title: Diagnostic and prognostic utility of mid-expiratory flow rate in older community-dwelling persons with respiratory symptoms, but without chronic obstructive pulmonary disease

Authors:

Gülmisal Güder (gueder_g@ukw.de)
Susanne Brenner (Brenner_S@ukw.de)
Stefan Störk (stoerk_s@ukw.de)
Matthias Held (matthias.held@missioklinik.de)
Berna DL Broekhuizen (B.D.L.Broekhuizen@umcutrecht.nl)
Jan-Willem J Lammers (J.W.J.Lammers@umcutrecht.nl)
Arno W Hoes (A.W.Hoes@umcutrecht.nl)
Frans H Rutten (F.H.Rutten@umcutrecht.nl)

Version: 2 Date: 23 March 2015

Author's response to reviews:

a